 Vc/11-Apr-04
Week 15  starting 5April 04

SM18 Operation Meeting (Tuesdays)

 Bench status  and tests for now & next few days : 

	Bench
	Name
	Status on Monday aft 5/04/04
	Temp

	TBB1
	MBAR1051
	PT 5.2 1.5 KA Quench HF
	1.9 K

	TBB2
	MBBL3122
	PT 13.2 Resist. Meas.
	300 K

	TBC1
	MBAR2016
	PT 12 Warm Up
	Warming

	TBC2
	SSS0013
	SSS_Provoked Quench
	1.9 K

	TBD1
	MBBL3118
	ICS 2 Connect Magnet (ICS)
	300 K

	TBD2
	MBAL1062
	PT 5.1 1.5 KA Quench LF
	1.9 K

	TBE1
	MBBL2081
	PT 10 Min. NRJ 11850 A
	1.9 K

	TBE2
	MBBL3086
	Prep 5 Pumping + Cool Down
	Cooling

	TBF1
	MBAL1057
	Prep 2.2 IAP @ warm LF
	300 K

	TBF2
	MBAL1061
	New Next      3111 ? 
	 (no AntiC)


EXPECTED NEW : 3119  ( no antic)
Test Coordinator this week: Marco B ; MTM support on call  :  (16-3602).

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Other General Issues
· Easter Break & ICS work
· Keep D1 free for SSS Mech Preparation ( 6 hrs ? )   J.Halik 
 [From 19 April…]
	Date
	Problem
	Type
	Application/soft
	Action/advise

	Tuesday 30_N

Wednesday 31_N


	TBE1: HV insulation @warm connected to CFB. No communication between the server and the megger.

TBD1: Same problem for HV cold.
	AB-CO
	HV
	2 mobiles rack were tried. Same problem. Federico called. He discovered that one switch in the RS 2323 C converter for the megger is in wrong position. 

Federico called and change switch position. Photo of the switch position pasted. 

The problem appeared again and the switch was moved to different position. 

	Wednesday 31_A

Thursday 01_A


	TBB: sun MTA42 does not allow sm 18 op account to access the storage of vol 8.
	AB/CO
	AQA
	Quench analysis done with sun mta 33.

Solved by Adrian.

	Wednesday 31_N


	TBB2 : Interlock “water” on PC
	MTM
	
	OK after closing other cluster’s valves.

	Thursday 01_M


	TBD1: For the SPA impossible to start the power: communication error.


	MTM


	
	Switch RS232 position error. Corrected by Federico.



	Thursday 01_N


	TBD2: IAP@warm Warning in the check of the sensor temperature TT821.
	MTM


	IAO


	Check of the connection cables.

But some warnings remain: NC open?

	Friday 02_M

Saturday_03_N
	TBF1: Training quench: Message in TEMA. “Test type could not be defined”. PLCSE setting error. But the test is defined!

Same problem on TBB2.
	AB/CO


	TEMA


	Test aborted and re-launched then OK.



	Friday 02_M

Friday 02_N


	TBB2: Acquisition problem. Differential signals saturated. Quench 5 where the four first cards signals were not present (D1-D2,…) 

Training quench 7: faults in AQA: signals are missing
	MTM/AB CO???


	HF,LF


	Federico called Adrian. Acquistion, cabling potaim-cards checked. Test re-launched and then OK for Q.6

Pierre will launch a deep investigation.



	Friday 02_N


	TBF1: HV test final at cold failed at MCS/ground.


	MTM


	HV


	Rack changed. Then OK.



	Saturday 03_M


	TBD1: Still differences in readings between TT821(301 K) and Tcryo (184 K).


	MTM/ACR


	
	Is the correct probe read? TT 150 A to be read between 30 K and 300 K.

TT148 between 30 K and 2 K.

TT149 below 2 K.



	Saturday 03_M 


	TBC1:IAP@cold. Launching test from TEMA no HF, LF graph were visible.

They were running locally.


	MTM


	HF, LF


	Applications closed and test re-launched.



	Saturday 03_M 


	TBC1: The channel 2 of the Tektronix used for IAP is faulty. 6 V measured even without signals (Ram)
	MTM


	
	????



	Saturday 03_M 


	TBB2: The access of the rotary switch is restricted by the presence of a pump. 
	ACR?


	
	ACR team informed Monday .



	Saturday 03_A 


	TBB2: Minimum energy. No charge of the power supply

	MTM


	
	Federico called.

1) control crate of the power supplies is not conform. Exchange with cluster A

2) the cable used to display the value of the charge has the wrong connector.

	Saturday 03_A 


	TBE1: HV test problem with rack 1 : error while putting the megger in remote mode and megger getting switch off automatically during calibration.


	MTM


	
	Federico called. Internal problem in the battery of the megger. Message: Power down because battery is too low. Federico send an e-mail to Richard. Test performed with another rack.



	Sunday 04_M 


	TBB2: Cryo team informs that there is an “ Anti cryostat fault”


	ACR/MTM?


	
	Federico informed and came.

Fast decrease in temperature of TA1C1 from +20C to -12 C during 30 minutes. Then temp increased again and reached the good value.



	Sunday 04_M 


	TBB2: Switch from B2 to CC: no lamp is glowing.
	MTM


	
	Federico informed. Many attempts. Ok after five times. 



	Sunday 04_A 


	TBB2: IAP@warm: Vtaps errors


	MTM


	IAP


	Federico came. IAP re-launched. But problem remained.

Giorgio forgot to restore two cables (link between potaim crates 1 and 3).Then OK (Jackie)








